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Bulletin　o／f　the　11？aculty　of　ll．ngineering，

H（）kka三do　IJnivers三ty，　No．85（】、977）

Heat　Transfer　in　a　Vertical　Glass　Wool　Layer

　　　　under　Constant　Heat　Flux　to　the　Wall

Nobuhiro　SEKI＊　Shoiehiro　Ful〈usAKO＊
　　　　　　　　　　　　Hideo　INABA＊

　　　　　　　　（Receivecl　March　31，　1977）

　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　Abstraet

　　　　This　report　deals　xKrith　the　effects　of　the　width　of　the　cavity　VV，　surface　temperature

o／f　ac　cold　wall　T，　and　the　specific　weight　of　glass　wool　r　on　apparent　heat　conductivity

2，．ガ（kcal／m　h　QC）or　tlユe　coe舐cient　of　heat　transferα（kcal／m2　h　OC）are　investigated　experi－

mentally　under　consta．nt　heat　flux　to　the　wall．

　　　　These　experimental　results　are　compared　with　the　previous　resultsi）　ttnder　isothermal

conditiolL　The　results　under　constant　heat　flux　show　a　decrease　of　5～15％ofλ、ガorα

as　compared　with　isothermal　condition　in　a　range　of　r＝’＝mO・w20　（kgfm3）．　（）n　the　other　hand，

i’n　a　range　of　7’＞20（kg／in3），　the　results　of　both　wail　heating　conditions　shovLT　almost　the

same　values　of　2eff　or　cv．

　　　　It　is　clear　that　the　value　of　λ、ガorα　is　in．dependellt　of　the　heating　condition　of　the

wall　in　a　range　of　7’＞20　（kg／m3）．

　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　1．　lntroduction

　　　　Glass　wool　insulating　material　is　widely　used　for　heat　insulation　of　buildings　etc．．

In　a　previous　reporti），　the　heat　transfer　phenomena　of　a　vertical　glass　wool　layer

uncler　isothermal　heating　of　the　wall　was　stuclied．　However，　in　industria｝　prac　ctice，

generally　numerous　heating　modes　of　constant　heat　fiux　to　the　wall，　for　example，

heat　accumulators　1ieated　by　sunheat　and　thermosyphon　heated　geothermal｝y．

　　　　Therefore，　to　clarify　the　influence　of　heating　conditions　of　the　wall　on　the　heat

transfer　in　a　glass　wool　layer　is　a　very　important　problem　in　practical　application．

　　　　This　experimental　stucly　investigates　the　effects　of　the　wiclth　of　the　cavity　’VI／，

the　cold　wall　surface　temperature　T，　and　the　specific　weight　of　glass　wool　r　on

apparent　heat　conductivity　with　natural　convective　heat　transfer　in　a　glass　wool　layer

under　constant　heat　flux　to　the　hot　wall　and　isothermal　colcl　wall，　namely，　mixed

bounclary　condition．

　　　　The　present　experimental　resul£s　ob£ained　are　correlated　with　the　apparent　heat

conductivity　7，，ff　or　the　coefllicient　of　heat　transfer　ct　The　present　experinients　are

carriecl　out　in　the　case　of　a　vertical　layer　packed　with　glass　wool　and　saturatecl

with　air　or　water　in　a　range　of　O〈r〈50　（1〈g／m3）．　The　results　of　the　present　work

are　discussed　in　comparison　to　previousi＞　results　uncler　isothermal　condition．

＊　Department　o／f　Mechanical　ILtngineering　II，　1’lokkaido　University，　Sapporo　06e，　Japan，
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　　　cl：

　　H：

　　vv：

　　　X：

　　　Y：

H／W：

　　7’ん：

　　7．も：

　　え砂∫：

　　　a’　：

　　　r：

　　　　　　　　　　　　　　　　　　　　　Nomenelature

heat　flux，　input　（kcal）fheating　surface　（in2）

height　of　cavity　（m）

width　of　cavity　（m）

distance　along　the　hot　wall　from　the　bottom　（m）

distance　perpendicular　to　the　hot　wall　（m）

aspect　ratlo

surface　temperature　of　the　hot　wall　（OC）

surface　temperature　of　the　cold　wall　（OC）

apparent　heat　conductivity　（kcalfm　h　OC）

coe舐cient　of　heat　transfer（kca1／m2　h。C）

specific　welght　of　glass　wool　（kg／m3）

　　　　　　　　　　　　　　　　　　　2．　Experimeittal　device　and　procedure

　　　　The　experimental　device　is　depicted　in　Fig．　1．　The　experimental　device　used

in　this　work　is　the　same　as　that　used　in　previous　worki），　namely，　the　hot　and　the

cold　walls　are　constructed　from　five　copper　plates　and　Cu－Co　thermo－couples　（O．3

mm　in　diameter）　are　embedded　in
　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　BR困E

each　copper　plate　to　obtain　the　sRw，E；

temperature　distributions　of　the

hot　and　the　cold　walls．

　　　　Three　cavities　having　a　vari－

able　width　of　W　（．？．2，　59　and　116

mm）　are　used　uncler　a　constant

height　H　（571　mm）．

　　　　In　order　to　compare　the　re一　1　j　V　l　1．．　i．．一！cFeoe‘T：：T

sults　obtained　under　both　wall

heatiBg　conditions，　the　heat　flux　q

to　the　hot　wall　is　selected　so　as　to　　h笥贈s蕊辱1。。⊂巴ゴ畠G　　　　　　　　・THE剛。⊂（xPEs①・。TwALし

。bt。i。　th。　mea。，u，face　t，mp。，a一　　　　　　　器鴫，。雛謎糊，、
　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　⑥ALUMINUM　PしATE
ture　of　the　hot　wall　Th．＝LtOOC．
　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　Fig．　1．　Schematic　diagram　of　the　experimental　device．
Surface　temperature　of　the　cold

wall　is　changed　from　100C　to　一150C．　Test　samples　of　the　glass　wool　guaranteed

by　the　Standard　of　the　JIS　A　9505　are　used．

　　　　The　experiments　are　carried　out　by　turning　on　the　selected　constant　electric

current　to　each　heater　after　packing　glass　wool　in　the　test　section．

　　　　Heat　transfer　is　measured　after　a　steady　state　in　the　section　is　reached　thermally

and　hydrodynamically．　ln　each　run，　it　requires　about　Lt　Nshrs　to　obtain　the　steady

state．　The　heat　loss　from　the　test　section　is　estimated　to　be　within　±3％　from

the　results　of　preliminary　test．
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　　　　　　　　　　　　　　　　　　　3．　Experimenta｝　results　and　discussions

　　　　a）　Temperature　distribution　of　the　hot　wall

　　　　First，　the　temperature　distributions　of　the　hot　wall　are　investigated　from　the

exlstence　of　a　temperature　gradient　in　a　vertical　direction　（X）　of　the　hot　wall　under

constant　heat飾x　may　be　cons呈dered　in　general　and　such　a　temperature　gradient

would　hcave　an　influence　on　the　heat　transfer　in　the　glass　woo｝　layer．

　　　　Typical　temperature　distributions　of

the　hot　wall　are　shown　in　Fig．　2　for

r＝10　（1〈g／m3）　together　wi£h　the　tem一　so

perature　distributions　shown　by　solid　A　40

1ines　in　case　of　isothermal　condition　S．30

（Tl，，＝．？．OOC）．　’‘　20
　　　　1n　Fig．2一（a）　of　water　as　the　work－
　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　10

ing　fluicl，　it　may　be　seen　that　with　the

declease　o．f　lhg．　tepiperaiuye．gradien．！　Otho　20　30　twt，（2．oc）30　rko　20　30

in　the　vertical　direction　of　the　hot　wall
　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　Fig．　2一（a），

along　with　the　clecreasiRg　width　IV，　the

heat　fiux　q　becomes　larger　in　the　case

of　mean　fixed　temperature　of　the　hot　so

wall．　Th．．＝＝P」OOC．　40
　　　　Fig．2一（b）　of　air　shows　the　same

qualitative　tendency　as　that　of　water，

but　the　temperature　gradient　of　air　in

cft　vertical　direction　is　smaller　than　that

of　water．　lt　can　be　understood　that

a　disparity　between　the　tendencies　of

these　temperature　gradients　occurs　not

only　due　to　the　smaller　viscosity　of　air

than　that　of　water，　but　also　because　of
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　　　　　　　　　　Fig，　2－Cb）．

Fig．　2．　Temperature　distribut’ions

　　　　　　　of　the　hot　xvalL

the　vertical　direction　（X＞　of　air

25

the正ength　of　the　boundary　layer　developing　in　　　　　　　　　　　　　　　　　　　　　　　　being

smaller　than　that　of　water．

　　　　b）　Tke　effeet　of　the　width　rv　of　the　cayity　on　A，ff　or　a

　　　　Fig．　3　shows　the　relation　between　the　width　V／　and　2，．tf　or　a’　for　vtu“iable　r　uncler

constant　heat　flux　together　with　that　under　isothermal　condition　indicated　by　solicl

i量nes．

　　　　Fig．　3一（a）　ancl　3一（b）　show　the　relation　between　W　and　2，ff　for　water　and　air　re－

spectively．　The　experlmental　resu｝ts　in　which　the　va｝ue　of　7，，fy　increases　as　1｝V　increases

are　obtained　under　both　wall　heating　conditions．　But，　the　values　of　20f」　under　constant

heat　flux　for　r＝O　and　10　（1〈g／m3）　show　a　clecrease　of　5　N　15％　as　compared　with　those

．しmder　isothermal　condition。

　　　　On　the　other　hand，　for　’x＝30　（kg／m3）　the　difference　of　the　values　of　1，ff　under

both　wall　heating　conditions　do　not　exist．　This　decrease　of　7，，tii　can　be　explainecl　by

the　reason　that　clue　to　the　existence　of　a　temperature　graclient　in　the　vertical
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　　Fig．3．　A　relation　between　the　wi〔lth　W　andλ8がorα。

direction　as　shown　in　Fig．　2，　the　increased　length　of　boundary　layer　in　the　present

experiments　occurs　and　the　value　of　2，if　representlng　the　mean　heat　transfer　in　the

glass　wool　layer　decreases　as　compared　with　that　under　isothermal　condition　without

a　temperature　gradient．

　　　　Fig．　3一（c）　and　3一（d）　illustrate　the　relation　between　W　and　cu　connected　with　2e．，Lf．

It　could　be　understood　from　these　figures　that　as　W　increases，　the　values　of　a’　de－

creases．　This　may　be　explained　by　the　following　reason．　The　term　of　ex　does　not

contain　implicitely　the　value　of　1］V．　Moreover，　as　W　increases　under　constant　height

（”＝：571　mm＞，　the　circulation　speed　of　fluid　becomes　weal〈　and　consequently　the　net

heat　transfer　from　the　hot　wall　to　the　cold　wall　decreases．

　　　　c）　The　effect　of　specifie　weight　of　glass　weol　T　on　A，ff　or　a

　　　　Fig．　4　illustrates　the　effect　of　the　specific　weight　“r　on　20jf　or　cr　for　variable　LV．

Fig．4一（a）and　4一（b）show　the　relation　betweenλ（ガandγfor　water　and　air　respec－

tively．　The　relation　between　P，，」7　and　r　under　both　heating　conditions　has　the　same

qualitative　tenclency　having　a　strong　naturac　l　convective　effect　in　a　range　of　7’〈20

（kg／m3）　．
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　　　　1n　a　range　of　small　W（22　and　59　mm），　the　values　ofλexf　under　cons£ant　heat

flux　cause　a　decrease　of　5～15％’as　compared　with　those　under　isothermal　condition

forγ〈20（kg／m2）．　On　the　other　hand，　a　d玉sparity　of　the　values　of　P、L，ff　uncler　both

wal｝heating　condltions　does　not　exist　forγ＞20（kg／m3）．　When　IV　becomes　l！6　mm，

the　values　of　7、eff　is　not　a｛lected　by　the　wall　heating　condition．　This　can　be　un．

derstood　that　the　wall　heat三ng　condition　does　not　affect　7、eff　because　of　the　weak：

natural　convection三n　the　layer　with　the　increasing．Vろ　On　the　contrary，　F圭g．4一（b）

for　W＝116　mm　using　a三r　shows　the　decrease　of？、ef．f　as　compared　with　that　for　iso．

thermal　condition．　This　may　be　explained　because　when　air　hav量ng　a　small　viscosity

is　used，　the　value　ofλ，ff　is　dependent　on　the　wall　heating　cond呈tion三n　a　wide　range

of皿
　　　　F1g．4（c）and　4一（d）illustrate　the　relation　betweenγandα．　It　can　be　seen　from

these丘gures　that　the　values　ofαdecreases　as　W　increases　due　to　the　reason　as

aforementioned　in　chapter（b）of　temperature　distribution．

　　　　d）The　effect　of　the　cold　wall　surface亡empera加re　Tc　onλeff

　　　　Fig．5shows£he　relation　between　T，　and？、eff　together　with　that　under　isother．

mal　condition　indicated　by　sohd　lines．

　　　　Fig．5一（a）and　5一（b）for　air　demonstrate　the　results　for　W；22　and　116　mm　re－

　　　　　0ユ5　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　Q35　　　　　FLUID：A【R，H・571mm．w・】16mrn
　　　　　　　　　　　　　FLVID．A［R．　H＝57】mm．W＝22mm

　　　て）　　　　　　　　　　　　　　　　　　　　　　　　　　　T　　　　　　　　　　＿v・T。〔kgim3）
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l㎝01∫二二二：：∵∵1σ30『》一・一“　・N’”一一一　

　　　べ
　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　一一　　1．τ．
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　　　　　　　　　　　　　　　　　　・1・10（kglm3＞、C。NSTANTHEAT　　　　　　　　　　・　30（、．）・・
　　　　　　　　　　　　　　　　　　　　　　　　　　　　FLUX
　　　　　　　　　　　　　　　　　　●　　　 30（　う　　），

　　　　　　。1・　’G　－1・、＝1－2・　　　。201。　　Q　－1。　．2。
　　　　　　　　　　　　　　　　　　Tc　（。c）　　　　　　．t．　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　Tc（。（＝）

　　　　　　　　　　　　　　　　　Fig．5一（a）．　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　Fig．5一（b）．

　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　FLU！DIWATER，H＝571mm，W＝116mm

　　　　1・5　FLUID：wATER’H＝57Tm「n　，w‘22mm　　　1．・／一一回目I・1…g1・3・
　　　　　　　　　　　　　　　　一　　1．下．

　　　δ　　　　　。r．IO（kg／m・）C．卜1．F．
　　　り　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　

　　　信　　　　　　　　　　．　　30（”　）　‘’　　　　　　　　　　9

　　　莞1・0　　　　　　　　　　　　　　　　　　　　　　　　　　　壱tO
　　　り　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　へ

誉／＼一一・3）　皇イー一㌃一．・．＿．3Gゆ，）
K・、5で’、、・一3・　・3・ @葱、5

　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　一一　　…．T．
　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　。1・IO｛k91m3）．C．旺

　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　●　　　　　　30　〔　　唱・　　　）・　　㌧

　　　　　01・一興ドS「一2・　　。1・㌦、。、一1・一2・

　　　　　　　　　　　　　　　　　Fig．5一（c）．　　　　　　　　　　　　　　　　　　　　　　　　　　Fig，5《d）．

　　　　　　　　　　Fig．5．　A　relation　between　the　cold　wal．1　surface　telnperature　1’c　and　2eff．
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spectively．　lt　can　be　said　from　these　figures　that　the　values　of　2，Jf　under　constant

heat　flux　show　a　decrease　of　5　N　15％　as　compared　with　those　unclex一　isothermal　con－

dition　for　r＝10　（kgfm3）　because　of　the　existence　of　the　temperature　graclient　in　the

vertical　direction．　On　the　other　hand，　for　r＝30　（kgfm3）　the　values　of　2，，i」　are　independ－

ent　of　the　heating　conciitions　of　wall　due　to　the　increased　resistance　to　a　moveinent

of　air　in　the　layer　with　the　increasing　r．

　　　　Typical　results　of　exl］eriments　for　water　are　shown　in　Fig．　5一（c）　and　5一（d）．　A

maximum　value　of　？，，f．f　exists　at　about　T，　＝50C　under　isothermal　condition，　while　the

existence　of　a　maximum　value　of　2，fr　is　not　clear　under　the　condition　of　constant

heat　flux．　This　can　be　explained　from　the　fact　that　temperature　clistribution　in　the

vertical　glass　wool　layer　is　affected　by　the　temperature　gradient　of　the　wall　in　the

vertical　direction，　that　is，　the　fiuid　layer　at　40C　distributes　comp｝icatedly　in　the　glass

wool　layer，　consequently，　the　effect　of　suppression　by　density　inversion　on　natural

convection　becomes　weal〈．

　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　4．　Conelusiolls

　　　　The　mechanism　of　heat　transfer　in　a　vertical　rectangular　cavity　packed　with

glass　wool　uncler　constant　heat　fiux　to　the　wall　is　investigated．

（1）　For　O〈r〈20　（kgfm3）　the　values　of　2，，．ff　under　constant　heat　flux　result　in　a　de－

crease　of　5”一u　15％　as　compared　with　those　under　isothermal　condition．　On　the　other

hand，　for　r＞20　（kg／m3）　the　value　of　2，f」・　is　independent　of　the　wall　heating　condition．

（2）　The　temperature　gradient　of　the　hot　wall　in　a　vertical　direction　induced　by　con－

stant　heat　flux　increases　as　H／　increases　under　the　present　constant　height　Fl，　and

this　tempera£ure　gradient　decreases　the　heat　transfer　in　the　glass　wool　layer．

（3）　The　net　heat　transfer　in　the　glass　wool　layer　is　clearly　shown　by　using　the

coe’fllicient　of　heat　transfer　cu　which　not　implicitly　contain　the　term　of　IV．
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